Author's response to reviews

Title: Moxibustion for treating knee osteoarthritis: study protocol of a multicentre randomised controlled trial

Authors:

Seunghoon Lee (aimforyou@hanmail.net)
Kun hyung Kim (pdchrist@empas.com)
Tae-Hun Kim (nejjing@kiom.re.kr)
Jung-Eun Kim (eujuki@kiom.re.kr)
Joo-Hee Kim (jhkim714@kiom.re.kr)
Kyung-Won Kang (kwkang@kiom.re.kr)
So-young Jung (syjung@kiom.re.kr)
Ae-Ran Kim (arkim@kiom.re.kr)
Hyo-ju Park (mable@kiom.re.kr)
Mi-Suk Shin (irisflower@kiom.re.kr)
Kwon-Eui Hong (hkeacu@dju.kr)
Ho-Sueb Song (hssong70@gachon.ac.kr)
Jin-Bong Choi (jy1907@gmail.com)
Hyung-Jun Kim (khjoongy@hanmail.net)
Sun-Mi Choi (smchoi@kiom.re.kr)

Version: 6 Date: 5 February 2013

Author's response to reviews: see over
Dear Editor-in-Chief,

We are pleased to resubmit the manuscript entitled "Moxibustion for treating knee osteoarthritis: study protocol of a multicentre randomised controlled trial" by Lee S at al.

We revised our manuscript according to reviewer. The details are as follows:

**Final editorial revisions**

1. Confirm that consent to publish the image featured in figure 3 was obtained from the subject of the photo. If so, please include this information in your manuscript. If not, then we would please ask you to remove this figure from your manuscript.

   **We changed the consented image featured in figure 3. And we added this information in Figure legend “A written consent for the picture was obtained from the subject. Photograph by Seunghoon Lee”. Two versions of consents (English and Korean) will be attached.**

2. Ensure that your revised manuscript conforms to the journal style (http://www.biomedcentral.com/info/ifora/medicine_journals ). It is important that your files are correctly formatted.

   **We conformed our manuscript to the journal style.**

We always thank to you in advance for your consideration of this manuscript.

Sincerely yours,

Sun-Mi Choi
Director/ Ph.D/ Doctor of Korean Medicine M.D. (DKM)/

Acupuncture, Moxibustion & Meridian Research Group

Medical Research Division